Author's response to reviews

Title: Collaborative development of an electronic Personal Health Record for people with severe and enduring mental health problems

Authors:

   Liam Ennis (liam.ennis@kcl.ac.uk)
   Daniel Robotham (Dan.Robotham@kcl.ac.uk)
   Mike Denis (MikeDenis01@gmail.com)
   Ninjeri Pandit (Ninjeri.Pandit@nhs.net)
   Dave Newton (David.Newton@slam.nhs.uk)
   Diana Rose (Diana.Rose@kcl.ac.uk)
   Til Wykes (Til.wykes@kcl.ac.uk)

Version: 2 Date: 14 April 2014

Author's response to reviews:

Line numbers have been inserted.